Author's response to reviews

Title: Different screening strategies (single or dual) for the diagnosis of suspected latent tuberculosis: a cost-effectiveness analysis

Authors:

Anil Pooran (anil.pooran@gmail.com)
Helen Booth (helen.booth@uclh.nhs.uk)
Rob F Miller (rmiller@gum.ucl.ac.uk)
Geoff Scott (geoff.scott@uclh.nhs.uk)
Motasim Badri (motasim.badri@uct.ac.za)
Jim F Huggett (j.huggett@ucl.ac.uk)
Graham Rook (g.rook@ucl.ac.uk)
Alimuddin Zumla (a.zumla@ucl.ac.uk)
Keertan Dheda (keertan.dheda@uct.ac.za)

Version: 5 Date: 5 February 2010

Author's response to reviews: see over
4th Feb 2010

To the Editor of BMC Pulmonary Disease

Thank you for accepting the manuscript 1685559911229299 entitled: “Different screening strategies (single or dual) for the diagnosis of suspected latent tuberculosis: a cost analysis”

All the necessary formatting changes have been made to the manuscript as requested. See format change request in bold followed by the change made in plain text.

Major revisions

Tables: table 5 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

Tables 5a and b has now been changed to table 5 and 6, respectively, and references to them in the text has also been changed accordingly

Minor revisions

Key words: please remove these

These have been removed

References: please provide the full citation details for reference 15, 25. You can view the BMC reference guide at the following link
http://www.biomedcentral.com/bmcgenet/ifora/#references. Please do not include DOI numbers in your reference list

Reference 25 has been changed to an updated version of the reference. The new reference is given as:

Health Protection Agency. Tuberculosis in the UK: Annual report on tuberculosis surveillance in the UK 2009.
[http://www.hpa.org.uk/web/HPAwebFile/HPAweb_C/1259152022594]
Also reference 15 is now updated and given as reference 11

Figures: Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.

This has now been changed in the text.

Tables- please remove the visible vertical lines from your tables

This has been removed from the tables.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been checked and all corrections have been made.